Author’s response to reviews

Title: Anti-inflammatory activities of astringent persimmons (Diospyros kaki Thunb.) stalks of various cultivar types based on the stages of maturity in the Gyeongnam province.

Authors:

Youngmin Kang (ymkang@kiom.re.kr)
Choi Jieun (dr98cju@naver.com)
Jung Haiyoung (haiyoung@kribb.re.kr)
Kim Mi jeong (tito006@kribb.re.kr)

Version: 2 Date: 28 Dec 2017

Author’s response to reviews:

We add Availability of data and materials in the manuscript.